Author's response to reviews

Title: Serratia marcescens outbreak in a neonatal intensive care unit: crucial role of implementing hand hygiene among external consultants.

Authors:

Carlotta Montagnani (carlotta.montagnani@unifi.it)
Priscilla Cocchi (priscilla.cocchi@unifi.it)
Laura Lega (l.lega@meyer.it)
Silvia Campana (s.campana@meyer.it)
Klaus P Biermann (k.biermann@meyer.it)
Cesare Braggion (c.braggion@meyer.it)
Patrizia Pecile (patrizia.pecile@unifi.it)
Elena Chiappini (elena.chiappini@unifi.it)
Maurizio de Martino (maurizio.demartino@unifi.it)
Luisa Galli (luisa.galli@unifi.it)

Version: 4  Date: 18 December 2014

Author's response to reviews: see over
Florence, 18th December 2014

Dear Editors,

Enclosed is the revised manuscript entitled:


We thank the editor and the reviewers for their favourable judgments.

We included a statement regarding ethical approval within methods section.

We believe that the manuscript has been much improved as a result of the revisions.

Sincerely yours,

Luisa Galli, MD
Infectious Disease Unit, Department of Health Sciences, University of Florence
viale Pieraccini, 24, I-50139 Florence, Italy.
phone: +39 055-5662439; fax: +39-055-4221012
e-mail: luisa.galli@unifi.it